Author’s response to reviews

Title: The value of episodic, intensive blood glucose monitoring in non-insulin treated persons with type 2 diabetes: Design of the Structured Testing Program (STeP) Study, a cluster-randomised, clinical trial [NCT00674986]

Authors:

William Polonsky (WHP@behavioraldiabetes.org)
Lawrence Fisher (fisherl@fcm.ucsf.edu)
Charles Schikman (endoctor2@hotmail.com)
Deborah Hinnen (DHinnen@madiabetesa.com)
Christopher Parkin (cgparkin@aol.com)
Zhihong Jelsovsky (zjelsovsky@biostatinc.com)
Linda Amstutz (linda.amstutz@roche.com)
Matthias Schweitzer (matthias_axel.schweitzer@roche.com)
Robin Wagner (robin.wagner@roche.com)

Version: 3 Date: 24 April 2010

Author’s response to reviews:

April 24, 2010

Dr Rikki Graham, PhD
BioMed Central Editorial Team

Dear Dr. Graham,

Thank you for your consideration of our manuscript and for the opportunity for publication in BMC Family Practice. The attached manuscript includes the revisions you requested. We have also made some minor corrections that were missed in the earlier version; these changes do not impact the content of the manuscript.

Please let me know if you have any questions or require additional information.

Sincerely,

William H. Polonsky, PhD
Principal Investigator
University of California, San Diego
San Diego, California, USA